Author's response to reviews

Title: Evaluating maternity care using national routine datasets: How are statistics affected by the quality of data on method of delivery?

Authors:

Hannah E Knight (hknight@rcog.org.uk)
Ipek Gurol-Urganci (ipek.gurol@lshtm.ac.uk)
Tahir A Mahmood (tmahmood@rcog.org.uk)
Allan Templeton (atempleton@rcog.org.uk)
David Richmond (d-richmond@rcog.org.uk)
Jan H van der Meulen (jan.vandermeulen@lshtm.ac.uk)
David A Cromwell (dcromwell@rcseng.ac.uk)

Version: 2 Date: 14 January 2013

Author's response to reviews:

A section on competing interests has been added to the manuscript.

Further details about the data used in the analysis have also been included in the acknowledgements section, as requested.